Author's response to reviews

Title: Regional differences in mortality in Greece (1984-2004): the case of Thrace

Authors:

Panagiotis PP Papastergiou (panpapast@med.uth.gr)
George GR Rachiotis (gsrachmed@yahoo.com)
Konstantina KP Polyzou (panpapast@med.uth.gr)
Christos CZ Zilidis (zilidis@gmail.com)
Christos CC Hadjichristodoulou (xhatzi@med.uth.gr)

Version: 7 Date: 15 August 2008

Author's response to reviews:

Dear Editorial production Team,

The list of authors has been written exactly as they are in the submission system.
The affiliation is in the following format: department, institute, city, and country.
We have changed the SUMMARY --> Abstract.
All unnecessary capitals have been removed.
Keywords have been removed.
The table title is now appear above the table and is now shorter.
Figures have been cropped.
All reference numbers are in square brackets.

Yours sincerely

Christos Hadjichristodoulou